Author's response to reviews

Title: Pretreatment with statins improves early outcome in patients with first-ever ischaemic stroke: a pleiotropic effect of statins or a beneficial effect of hypercholesterolemia?

Authors:

Adrià Arboix (aarboix@hscor.com)
Luís García-Eroles (lgarciaer@csdm.es)
Montserrat Oliveres (11466moi@comb.es)
Cecília Targa (ctargab@meditex.es)
Miquel Balcells (mibalcells@hotmail.com)
Joan Massons (jmassonsc@terra.es)

Version: 4 Date: 18 June 2010

Author's response to reviews:

We have included: 1)In the first page: the authors initials; 2)In the Authors’ contributions (page 11): The authors initials and the sentence "All have read and approved the final draft" 3)The funding information under acknowledgement: 4)key words removed.

Thank you very much for your attention to this submission.

Yours sincerely,

Adrià Arboix, MD, PhD